Author’s response to reviews

Title: Health mediators as members of multidisciplinary group practice: lessons learned from a primary health care model programme in Hungary

Authors:
Karolina Kósa (kosa.karolina@sph.unideb.hu)
Cintia Katona (katona.cintia@sph.unideb.hu)
Magor Papp (magorpapp@gmail.com)
Gergely Fürjes (furjesgergely@gmail.com)
János Sándor (sandor.janos@sph.unideb.hu)
Klára Bíró (kbiro@med.unideb.hu)
Róza Ádány (adany.roza@sph.unideb.hu)

Version: 1 Date: 17 Nov 2019

Author’s response to reviews:
Dear Dr. Collins,

Thank you very much for the Editor's comments.

We re-composed the abstract adding information regarding health mediators.
The section on health mediators was moved from the Discussion to Background.
We extended the Methods referring to the mandatory data collection of the Programme.

Comparison of findings was moved from Results to the Discussion.

We rephrased the last paragraph of Results to make it clear.

The section on external validity was deleted, Conclusion was re-composed.

English language editing was performed.

We hope you will now find the manuscript worthy of peer review.

Best regards,
Karolina Kósa